Author's response to reviews

Title: Abnormal Urinary Excretion of NKCC2 and AQP2 in Response to Hypertonic Saline in Chronic Kidney Disease. An Intervention Study in Patients with Chronic Kidney Disease and Healthy Controls

Authors:

Janni M Jensen (jannimaj@gmail.com)
Frank H Mose (Frank.Holden.Christensen@vest.rm.dk)
Anna-Ewa O Kulik (Anna.Ewa.Oczachowska-kulik@vest.rm.dk)
Jesper N Bech (Jesper.Noergaard.Bech@vest.rm.dk)
Robert A Fenton (ROFE@ANA.AU.DK)
Erling B Pedersen (Erling.Bjerregaard.Pedersen@vest.rm.dk)

Version: 3
Date: 30 April 2014

Author's response to reviews:

Holstebro, 30.04.2014

Dear Editors of BMJ Nephrology

Thank you for your reply regarding our submitted manuscript titled “Abnormal Urinary Excretion of NKCC2 and AQP2 in Response to Hypertonic Saline in Chronic Kidney Disease. An Intervention Study in Patients with Chronic Kidney Disease”, reference number MS: 1908766592122614

We have now addressed the comments raised by the reviewers and reviewed the manuscript. A point to point-by-point response has been made to each of the two reviewers comments. Our response is included in a separately attached file. Within the revised manuscript, deleted passages are specified in the margin, and added text is highlighted with red. As an editorial request, emails of all the authors have been added to the title page.

I’m looking forward to hearing from you again.

Kind Regards

Janni Majgaard Jensen, M.D.
Departments of Medical Research and Medicine
Holstebro Hospital
Laegaardvej 12
7500 Holstebro
Denmark
Tel no: +45 7843 6588
Fax no: +45 7845 6582
Email: jannimaj@gmail.com